Author's response to reviews

Title: Symptom attributions in patients with colorectal cancer: a population based study

Authors:

Line F Jensen (line.jensen@ph.au.dk)
Line Hvidberg (line.hvidberg@ph.au.dk)
Anette F Pedersen (afp@feap.dk)
Peter Vedsted (p.vedsted@feap.dk)

Version: 2 Date: 9 March 2015

Author's response to reviews: see over
Dear editor,

Thank you for the opportunity to make changes to the format of the article. The specified changes has now been incorporated into the manuscript including:

- Numbers have been included on each page and line numbers has been included as well.
- We have included a list of abbreviations after the conclusion
- The Competing Interest section has been moved before Authors Contribution
- The Acknowledgement section has been moved after Authors Contribution
- The tables is now presented after the References
- A section on ethical considerations is now presented in the method section

We hope that this will be sufficient to send the manuscript in review. If you have other comments please do not hesitate to write again.

Yours sincerely (on behalf of the authors),

PhD Fellow, Line Flytkjær Jensen

Research Centre for Cancer Diagnosis in Primary Care (CaP), Research Unit for General Practice, Aarhus University, Denmark.